Author's response to reviews

Title: Acupuncture Increases the Total Neuron Number in the Hippocampal CA1 Area of Rats with Multi-infarction Dementia

Authors:

Fang Li (m13663420179@163.com)
Chao-Qun Yan (857620507@qq.com)
Li-Ting Lin (1065286401@qq.com)
Hui Li (lihuilihui1314@163.com)
Xiang-Hong Zeng (806656628@qq.com)
Yi Liu (913418574@qq.com)
Si-Qi Du (dsq1992@yeah.net)
Wen Zhu (zhuwen56625658@126.com)
Cun-Zhi Liu (lcz623780@126.com)

Version: 2
Date: 18 November 2014

Author's response to reviews:

Dear editor,

In the process of "author details", the name of author 1 is wrong, we have revised "Fang Li Li" to "Fang Li", but in the manuscript, we did not have any revision.

Sincerely yours,
Cun-Zhi Liu